Math 221 Review sheet for Test 2 Spring 2001

1. NO CALCULATORS ALLOWED

2. Test 1 covers the Sections 4.1-4.8, 5.1 and 8.4. Anything done in class is fair game for the test.

3. Theory is 30%: One or two of the following proofs will be on the test:

Theorems 4.2, 4.3, 4.5, and 4.10; problems 4.6 T8, 4.3 T6, 5.1 T12. A short proof that you have not seen may also be on the test.

4. The rest of the test (70%) involves calculations. Of course understanding the theory can make some calculations easier.